\documentclass[12pt]{article}
\usepackage{fullpage,graphicx}

\title{Calculus Lab, Week 3}
\author{Your Name Here}

\begin{document}

\maketitle


Our goal this week is to answer some mathematical questions using Maple,
then write up our solutions in LaTeX.  For the lab, write a LaTeX report
with three sections (one section for each question below).  

Before next week's lab, upload the tex and PDF files to a new CLEO folder
for the week.  You do not have to turn in the Maple worksheet.  There's 
a template LaTeX file you can use on the class website to get started.

\section{Exercise:  Riemann Sums}



\section{Exercise:  Find the Pattern}


\section{Exercise: More on Plotting}

Here, include the plot created in Maple.

\end{document}